Author's response to reviews

Title: Reliability and cultural applicability of the Greek version of the International Personality Disorders Examination.

Authors:

Konstantinos N Fountoulakis (kfount@med.auth.gr)
Apostolos Iacovides (kfount@med.auth.gr)
Christina Ioannidou (kfount@med.auth.gr)
Francesca Bascialla (kfount@med.auth.gr)
Ioannis Nimatoudis (nimatoud@med.auth.gr)
George St Kaprinis (kaprinis@med.auth.gr)
Aleksandar Janca (ajanca@cyllene.uwa.edu.au)
Alv A Dahl (a.a.dahl@psykiatri.uio.no)

Version: 5 Date: 14 May 2002

Dear Editor

We made the corrections needed in the title and in the format of the manuscript. Four separate files will be uploaded substituting the final pages of the original manuscript you said you are unable to reproduce.
I think that the paper is in accord with your guidelines

Best results

Dr KN Fountoulakis